Author's response to reviews

Title: Cost analysis and effectiveness of one-stage laparoscopic versus two-stage endolaparoscopic management of cholecystocholedocholithiasis: a retrospective cohort study

Authors:

Anne Mattila (anne.mattila@ksshp.fi)
Johanna Mrena (johanna.mrena@ksshp.fi)
Ilmo Kellokumpu (ilmo.kellokumpu@ksshp.fi)

Version: 4 Date: 28 Jun 2017

Author’s response to reviews:

Detailed response to Editorial Office

Dear Helen Armes,

following amendments have been made to the manuscript with red highlighting:

1. The study was approved by Ethics Committee of the Central Hospital of Central Finland.

We hope that the corrections made in the manuscript will achieve your acceptance

On behalf of the writing team

Anne Mattila, M.D.